Author's response to reviews

Title: Long Term follow up (10 years) after En-Bloc Resection and Reconstruction of a Solitary Paraganglioma Metastasis in the first lumbar vertebral body. A Case Report

Authors:

Alexander Richter (al.richter@asklepios.com)
Henry Halm (HHalm@Schoen-Kliniken.de)
Ulf Liljenqvist (ulf.liljenquist@sfh-muenster.de)
Thomas Lerner (thomas.lerner@sfh-muenster.de)
Markus Quante (MQuante@Schoen-Klinken.de)

Version: 6 Date: 28 May 2010

Author's response to reviews:

reduce the number of references to 30.
include the ethnicity of the patient